Author's response to reviews

Title: The changing landscape of care: does ethics education have a new role to play in health practice?

Authors:

Julie Wintrup (J.Wintrup@soton.ac.uk)

Version: 5 Date: 11 February 2015

Author's response to reviews:

Julie Wintrup
The changing landscape of care: does ethics education have a new role to play in health practice?

- Competing interests
I have no competing interests

ON 11TH FEB 2015 I added the following to the manuscript:
Competing interests
The author declares that she has no competing interests.

Author's contribution
As the sole author of the paper JW was responsible its conception, design, drafting and writing. She is accountable for all aspects of the work including its accuracy and integrity.

Julie Wintrup
11TH FEB 2015